Author's response to reviews

Title: Controversies in the treatment of invasive urothelial carcinoma: a case report and review of the literature

Authors:

Vicente Guillem (vguillem@fivo.org)
Miguel A Climent (maclimnet@fivo.org)
Javier Cassinello (jacaes@sescam.jccm.es)
Emilio Esteban (eesteban@seom.org)
Daniel Castellano (cdanicas@hotmail.com)
José L Gonzalez Larriba (jlgonzalezlarriba@telefonica.net)
Pablo Maroto (jmaroto@santpau.cat)
Carlos Camps (camps_car@gva.es)

Version: 2 Date: 26 May 2014

Author's response to reviews:

Dear editor,

According to the editor suggestions, we have added an Acknowledgment section (between the Authors’ contributions section and Reference list). In the new attached document we include line numbers to allow both, the editorial team and the referees, to assess our manuscript more efficiently.